Author's response to reviews

Title: Effectiveness of physiotherapy exercise following total knee replacement: systematic review and meta-analysis

Authors:

Neil Artz (neil.artz@plymouth.ac.uk)
Karen Elvers (Karen.Elvers@bristol.ac.uk)
Catherine MinnsLowe (Catherine.MinnsLowe@ouh.nhs.uk)
Catherine Sackley (catherine.sackley@kcl.ac.uk)
Paul Jepson (p.d.jepson@bham.ac.uk)
Andrew Beswick (Andy.Beswick@bristol.ac.uk)

Version: 2 Date: 17 September 2014

Author's response to reviews: see over
RE: changes made to manuscript.

Dear editorial team,

Thank you for your email requesting changes to the submission.

The following changes have been made;

1. Page numbers have been added to the manuscript.
2. Line numbers have been added to the manuscript.

Regarding the Ethics and Consent statements requested, the manuscript reports a systematic review of the literature. No research on human subjects were required therefore research ethical and consent statements are not applicable.

Yours sincerely

Dr Neil Artz PhD
Lecturer in Physiotherapy
University of Plymouth